Author's response to reviews

Title: Investigating socio-economic-demographic determinants of tobacco use in Rawalpindi, Pakistan

Authors:

Ali Alam (aliyawaralam@gmail.com)
Azhar Iqbal (azhar@heartfile.org)
Khalif Bilr Mohamud (drbile@who.pa.org)
Ronld Laporte (ronlaporte@aol.com)
Ashfaq Ahmed (ddgih@hotmail.com)
Sania Nishtar (sania@heartfile.org)

Version: 5 Date: 6 February 2008

Author's response to reviews: see over
Investigating socio-economic-demographic determinants of tobacco use in Rawalpindi, Pakistan

(Revised Cover Letter)

Ali Yawar Alam¹, Azhar Iqbal², Khalif Bile Mohamud³, Ronald E Laporte⁴, Ashfaq Ahmed⁵, Sania Nishtar⁶
Shifa college of Medicine, Islamabad, Pakistan, Heartfile, W.H.O & Ministry of Health Islamabad, Pakistan

1. Assistant Professor, Community Health Sciences, Shifa College of Medicine, Islamabad, Pakistan.
2. Program Officer, Heartfile, Islamabad, Pakistan
3. W.H.O Representative Pakistan
4. Director Disease Monitoring and Telecommunications WHO Collaborating Centre, Professor of Epidemiology, University of Pittsburgh, U.S.A
5. Ministry of Health, Islamabad, Pakistan
6. Founder and President, Heartfile, Islamabad, Pakistan

Why BMC Public Health should publish this Research article:
Tobacco control is an important area of public health worldwide, more so in the developing world where awareness levels regarding their potential for harm are very low. This original Research work conducted in a developing country setting needs a wider audience in other such nations where tobacco consumption and its sequale have wide range health implications. In my view BMC public health is just the right place to publish this manuscript, because of it’s speedy process, online availability and wide audience.

Corresponding Author

Ali Yawar Alam, MBBS, MS(U.K) MPH(U.S.A)
Assistant Professor, Community Health Sciences, Shifa College of Medicine, Islamabad, Pakistan

Email: alyayaralam@gmail.com
Telephone: 92-51-4603751
Fax: 92-51-4435046

Ethics Committee Clearance:
Ethical clearance for this study was obtained from ‘Ethics committee at Heartfile, Islamabad, Pakistan,’ in September 2004. Informed consent was obtained from the study participants.
POINT BY POINT RESPONSES

Formatting changes requested
---------------------------------------
Please remove the repeated title from above the abstract.  **DONE**

Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: [http://www.biomedcentral.com/info/ifora/figures](http://www.biomedcentral.com/info/ifora/figures)  **DONE**

References- references 11 and 12 are mentioned out of order. Please cite the references in order. Although all effort will be made to appropriately place the table within the text, for referencing purposes, please assume the tables come at the end of the manuscript.

  REFERENCES 11 AND 12 HAVE BEEN CITED IN TABLE 1, WHICH IS PART OF METHODS

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system. **DONE**

We have not been able to contact Khalif Bile Mohamud using this address in the past. Please verify this address or provide an alternate address.

  ALTERNATE EMAIL ADDRESS ADDED:  BILEK@pak.emro.who.int

Thank You

Dr. Ali Yawar Alam
aliyawaralam@gmail.com

6th Feb, 2008